vc/11-Nov-03
Week 46 starting 10 nov 03

SM18 Operation SCHEDULE Meeting (Tuesdays)

Schedule for this week : 
	BENCH
	CURRENT

MAGNET
	STATUS 

Today(11/11/03)
	Forward look,   & Next Magnet 

	TBA1
	NOT Active
	
	Bench de-activated – preparation for D bench

“Park” here 2001 ( moved from D bench )

	TBA2
	3055
	warming
	warm@2400 hrs 11/11 2029 next ?or thermal cycle

	TBC1
	1048
	To disconnect
	3027 next ?

	TBC2
	SSS3
	cold
	

	TBE1
	1034
	connection
	Start tests after E2 finished

	TBE2
	3008
	Cooling?
	cold@22hrs 11/11; Start tests when cold

	TBF1
	1021
	Cold tests
	

	TBF2
	1039
	Warm 
	Was warm 10/11 1800hr 3022 next ?


 In pipeline Next 3034

Test Coordinator this week:  nikolay;  MTM  Tech. on call:  (16-3602) : 

· Situation From Cryogenics:  J-P Lamboy or ACR Rep.

· Other General Issues
1. New Instructions: Joints, RRR, Q table u files filling, GHH Sequences /times

2. new todo list  numbers…

3. AOB

SnapShot of STATUS NOW(Tues 11/11am):

	SM18 LHC Magnet Test Facility [image: image1]


	TBA1
	_
	_
	 


	TBA2
	MBAR3055
	PT12 Warm Up
	Warming


	TBB1
	_
	_
	 


	TBB2
	_
	_
	 


	TBC1
	MBBL1048
	PT12 Warm Up
	300 K


	TBC2
	SSS004
	Prep2 HV Test Warm
	300 K


	TBD1
	_
	_
	 


	TBD2
	_
	_
	 


	TBE1
	MBBR3012
	Prep3 HV Initial @ Warm
	300 K


	TBE2
	MBBR1034
	Prep2.3 IAP @ warm LF
	300 K


	TBF1
	MBBR1021
	MM2 LHC Cycle
	1.9 K


	TBF2
	MBBR1039
	PT13.1 IAP @ Warm
	300 K


[image: image2.png]



Current (pm) Bench Status: 

	Date
	Problem
	Type
	Application/soft.
	Action/advise

	Tuesday 04_N


	TBE2: Quench in Vclamp

Problem already appeared in the morning shift.


	MTM
	
	Cryo says that everything is OK

Exchange of V clamp card with that of TBF2

Origin: Vtap wire disordered in the CFU external to connector.

	Wednesday 05_M


	TBC1: TT821 : no reading because Tsensor coefficients not present in MTF


	ACR
	MTF
	

	Thursday 06_A


	TBC1: Tcoil : switch in the mobile rack is not changing from MM to Tcoil

D1_S09_C opened.
	MTM
	
	PLC not properly configured?

 

	Thursday 06_N


	TBC1: Provoked quench all LF :

Problem in configuration file
	MTM
	
	 

	Friday 07_M


	TBF2: Temp probe broken
	ACR
	
	 

	Friday 07_A


	TBF2: MM0 problem with end stops
	MTM
	
	Solved by Ewald.

 

	Friday 07_A


	TBF2: No test type for MM position search included in the test sequence.
	MTM
	TEMA
	Solved by Maryline.

 

	Friday 07_A


	TBF2: Quench at 1.5 kA (all LF)

Extension of quench not properly set by TEMA
	MTM
	
	Quench repeated

 

	Friday 07_A


	TBF2: Quench all HF

Cryo OK lost while ramping
	ACR
	
	 

	Friday 07_A


	TBA2: IAP at cold Tcoils.

One channel shorted anther shows 431 K
	MTM
	
	 

	Friday 07_A


	TBC1: training #4

SPA ? Interlock glowing.

No trigger, no quench.
	MTM/ACR
	
	Cryo OK lost ?

 

	Saturday 08_M


	TBC1: Full load line /ext. joint measurements

Message”one bad measurement removed”
	MTM
	Joint measurement.
	Bug in the software

Solved by Olav.

 

	Saturday 08_M


	TBC1: HV at cold 

Ext&int. Bus Bar v/s ground failed. Test with rack 3.
	MTM
	
	Re-launched with another rack (rack 2) and OK

 

	Saturday 08_A


	TBF2: Min energy not performed because Cryo_X2 signal not OK.
	ACR
	
	 

	Saturday 08_A


	TBF2: Min energy at 11850 A: Configuration file not found..

LF& HF crates switch on/off : MXI error.


	MTM
	
	LF& HF crates reset. Then OK.

 

	Saturday 08_N


	TBE2: QH always on when the transparent panel is removed..


	MTM
	
	QH voltage not below 50 V.

Reset and discharge them.

 

	Saturday 08_N


	TBF2: Cryo-lost during detraining check.


	ACR
	
	 

	Sunday 09_M


	TBF2: LHC cycle

Program stopped after the cleansing cycle.

“Motor end stop and encoder error”.

CPU error when receiving the pulse.


	MTM
	
	Switch off the power supply of the integrators for 10 mms.

Problem remaining

 

	Sunday 09_M


	TBF2: LHC cycle

During a ramp rate current measured not readable.


	MTM
	MMP
	Done on purpose during the ramps of current.

 

	Sunday 09_M


	TBF1: Settings MMP to be done


	MTM
	MMP
	TBF1: Settings MMP updated by Maryline.

 

	Sunday 09_M


	TBF2: VME error while testing integrators.

Integrator cards of the 3rd crate are faulty.


	MTM
	
	Check by Ewald.

 

	Sunday 09_M


	TBC1: IAP @warm final , error found in TT821.


	MTM
	
	No temperature sensor in the magnet of TBC1?

 

	Sunday 09_M


	TBF2: Final HV insulation @warm : change of the switch (CC to F2) Position.. Switch blocked. 


	MTM
	
	Maryline carried out the change.

 

	Sunday 09_N


	TBA2: After MM: MMP is hung 


	AB/CO
	MMP
	Reboot the sun to continue

 

	Mon 10_M


	TBF1: MM position search 

MUX , VME, motor errors.


	MTM
	
	Test aborted.

 


Problems of Last week

	Date
	Problem
	Type
	Application/soft.
	Action/advise

	Tuesday 04_N


	TBE2: Quench in Vclamp

Problem already appeared in the morning shift.


	MTM
	
	Cryo says that everything is OK

Exchange of V clamp card with that of TBF2

Origin: Vtap wire disordered in the CFU external to connector.

	Wednesday 05_M


	TBC1: TT821 : no reading because Tsensor coefficients not present in MTF


	ACR
	MTF
	

	Thursday 06_A


	TBC1: Tcoil : switch in the mobile rack is not changing from MM to Tcoil

D1_S09_C opened.
	MTM
	
	PLC not properly configured?

 

	Thursday 06_N


	TBC1: Provoked quench all LF :

Problem in configuration file
	MTM
	
	 

	Friday 07_M


	TBF2: Temp probe broken
	ACR
	
	 

	Friday 07_A


	TBF2: MM0 problem with end stops
	MTM
	
	Solved by Ewald.

 

	Friday 07_A


	TBF2: No test type for MM position search under the test  sequence.
	MTM
	TEMA
	Solved by Maryline.

 

	Friday 07_A


	TBF2: Quench at 1.5 kA (all LF)

Extension of quench not properly set by TEMA
	MTM
	
	Quench repeated

 

	Friday 07_A


	TBF2: Quench all HF

Cryo OK lost while ramping
	ACR
	
	 

	Friday 07_A


	TBA2: IAP at cold Tcoils.

One channel shorted anther shows 431 K
	MTM
	
	 

	Friday 07_A


	TBC1: training #4

SPA ? Interlock glowing.

No trigger, no quench.
	MTM/ACR
	
	Cryo OK lost ?

 

	Saturday 08_M


	TBC1: Full load line /ext. joint measurements

Message”one bad measurement removed”
	MTM
	Joint measurement.
	Bug in the software

Solved by Olav.

 

	Saturday 08_M


	TBC1: HV at cold 

Ext&int. Bus Bar v/s ground failed. Test with rack 3.
	MTM
	
	Re-launched with another rack (rack 2) and OK

 

	Saturday 08_A


	TBF2: Min energy not performed because Cryo_X2 signal not OK.
	ACR
	
	 

	Saturday 08_A


	TBF2: Min energy at 11850 A: Configuration file not found..

LF& HF crates switch on/off : MXI error.


	MTM
	
	LF& HF crates reset. Then OK.

 

	Saturday 08_N


	TBE2: QH always on when the transparent panel is removed..


	MTM
	
	QH voltage not below 50 V.

Reset and discharge them.

 

	Saturday 08_N


	TBF2: Cryo-lost during detraining check.


	ACR
	
	 

	Sunday 09_M


	TBF2: LHC cycle

Program stopped after the cleansing cycle.

“Motor end stop and encoder error”.

CPU error when receiving the pulse.


	MTM
	
	Switch off the power supply of the integrators for 10 mms.

Problem remaining

 

	Sunday 09_M


	TBF2: LHC cycle

During a ramp rate current measured not readable.


	MTM
	MMP
	Done on purpose during the ramps of current.

 

	Sunday 09_M


	TBF1: Settings MMP to be done


	MTM
	MMP
	TBF1: Settings MMP updated by Maryline.

 

	Sunday 09_M


	TBF2: VME error while testing integrators.

Integrator cards of the 3rd crate are faulty


	MTM
	
	Check by Ewald.

 

	Sunday 09_M


	TBC1: IAP @warm final , error found in TT821.


	MTM
	
	No temperature sensor in the magnet of TBC1?

 

	Sunday 09_M


	TBF2: Final HV insulation @warm : change of the switch (CC to F2) Position.. Switch blocked. 


	MTM
	
	Maryline carried out the change.

 

	Sunday 09_N


	TBA2: After MM: MMP is hung 


	MTM
	MMP
	Reboot the sun to continue

 

	Mon 10_M


	TBF1: MM position search 

MUX , VME, motor errors.


	MTM
	
	Test aborted.

 


